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NuMI Update #

7

® Water and air-flow samples taken during Tuesday and Thursday
shutdowns to help determine the sources of the tritium - data
are still being analyzed

e Target hall air cooling chiller tripped possibly because of a leaky
seal - no beam down time

e Contractors are working on the problem
® Beam can still run while the work is being done
e Started low intensity running today and will run for 3 days

e ~2el2 POT/spill

® Done to quantify beam rate systematics
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Beam Delivered Since May 1, 2005

Daily Protons Delivered to NuM I Target Starting1 May 2005

Protons Delivered on Date (x E17)
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MINOS Status #

7

® Far Detector
e Uptime still near 100%

e | PMT that has been showing low rates for a while was replaced last
week

® Several other problem channels in one crate were diagnosed and
fixed as well

® Near Detector
e Uptime at >99% for the week

® 2 bad channels swapped
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Detector Performance

Number of POTs vs Time
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3 Reconstructed Events and POT vs T|me L, .

E wns ams mms 11f[15 12m5
40000 —
= —
0 — . .
E 300005 .- r .J:..‘. M 'j.
c — - R -
@D 20000__ & .' 1 - -
IJ:J = - L * _ﬁ.‘; :.-:i*
10000 VA ogte
- A <
: =' | N
0o 100
Days Since May 1, 2005
Post Trltlum Shutdowns
50 ————————— —
= 5/05 6/05 7/05 8/05 9/05 mms 11f[15 12m5
£ 40—
[ —
g Far
5 F { |
5 20p | { _
o i ; }
O 50 100 150 200

1/06

L |

Days Since May 1, 2005

250

2

=k
n

—l

=
in

Lad
=

M2
=

=k
=

#

10" POT/Day

\ Post Tevatron

Shutdown

10"® POT/Month

® All neutrino events (CC and NC) passing minimal cuts are shown

® Number of observed events follows POTs quite well

® Far detector data from the open set
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Event Rate vs Time

Horn Current Tests

Post Tritium Shutdowns

Days Since May 1, 2005

® Event rate is quite stable over time
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